# Models estimating risk of hepatocellular carcinoma in patients with alcohol or NAFLD-related cirrhosis for risk stratification

# Graphical abstract



# Highlights

- HCC risk varies dramatically in patients with cirrhosis.
- We developed models estimating HCC risk in patients with NAFLD-cirrhosis or ALD-cirrhosis.
- The models use simple, readily available predictors.
- The models are available as web-based tools at www. hccrisk.com.

# Authors

George N. Ioannou, Pamela Green, Kathleen F. Kerr, Kristin Berry

# Correspondence

georgei@medicine.washington.edu (G.N. Ioannou)

# Lay summary

Patients with cirrhosis of the liver are at risk of getting hepatocellular carcinoma (HCC or liver cancer) and therefore it is recommended that they undergo surveillance for HCC. However, the risk of HCC varies dramatically in patients with cirrhosis, which has implications on if and how patients get surveillance, how providers counsel patients about the need for surveillance, and how healthcare systems approach and prioritize surveillance. We used readily available predictors to develop models estimating HCC risk in patients with cirrhosis, which are available as web-based tools at www.hccrisk.com.



# Models estimating risk of hepatocellular carcinoma in patients with alcohol or NAFLD-related cirrhosis for risk stratification

George N. Ioannou<sup>1,2,3,\*</sup>, Pamela Green<sup>3</sup>, Kathleen F. Kerr<sup>4</sup>, Kristin Berry<sup>3</sup>

<sup>1</sup>Division of Gastroenterology, Veterans Affairs Puget Sound Healthcare System and University of Washington, Seattle, WA, United States; <sup>2</sup>Department of Medicine, Veterans Affairs Puget Sound Healthcare System and University of Washington, Seattle, WA, United States; <sup>3</sup>Research and Development, Veterans Affairs Puget Sound Healthcare System, Seattle, WA, United States; <sup>4</sup>Department of Biostatistics, University of Washington, Seattle, WA, United States

**Background & Aims**: Hepatocellular carcinoma (HCC) risk varies dramatically in patients with cirrhosis according to welldescribed, readily available predictors. We aimed to develop simple models estimating HCC risk in patients with alcoholrelated liver disease (ALD)-cirrhosis or non-alcoholic fatty liver disease (NAFLD)-cirrhosis and calculate the net benefit that would be derived by implementing HCC surveillance strategies based on HCC risk as predicted by our models.

**Methods**: We identified 7,068 patients with NAFLD-cirrhosis and 16,175 with ALD-cirrhosis who received care in the Veterans Affairs (VA) healthcare system in 2012. We retrospectively followed them for the development of incident HCC until January 2018. We used Cox proportional hazards regression to develop and internally validate models predicting HCC risk using baseline characteristics at entry into the cohort in 2012. We plotted decision curves of net benefit against HCC screening thresholds.

**Results**: We identified 1,278 incident cases of HCC during a mean follow-up period of 3.7 years. Mean annualized HCC incidence was 1.56% in NAFLD-cirrhosis and 1.44% in ALD-cirrhosis. The final models estimating HCC were developed separately for NAFLD-cirrhosis and ALD-cirrhosis and included 7 predictors: age, gender, diabetes, body mass index, platelet count, serum albumin and aspartate aminotransferase to  $\sqrt{alanine}$  aminotransferase ratio. The models exhibited very good measures of discrimination and calibration and an area under the receiver operating characteristic curve of 0.75 for NAFLD-cirrhosis and 0.76 for ALD-cirrhosis. Decision curves showed higher standardized net benefit of risk-based screening using our prediction models compared to the screen-all approach.

**Conclusions**: We developed simple models estimating HCC risk in patients with NAFLD-cirrhosis or ALD-cirrhosis, which are available as web-based tools (www.hccrisk.com). Risk stratification can be used to inform risk-based HCC surveillance strategies in individual patients or healthcare systems or to identify high-risk patients for clinical trials. **Lay summary**: Patients with cirrhosis of the liver are at risk of getting hepatocellular carcinoma (HCC or liver cancer) and therefore it is recommended that they undergo surveillance for HCC. However, the risk of HCC varies dramatically in patients with cirrhosis, which has implications on if and how patients get surveillance, how providers counsel patients about the need for surveillance. We used readily available predictors to develop models estimating HCC risk in patients with cirrhosis, which are available as web-based tools at www.hccrisk.com. © 2019 European Association for the Study of the Liver. Published by Elsevier B.V. All rights reserved.

## Introduction

Annual HCC risk varies greatly in patients with cirrhosis ranging from as little as <0.2% to as high as >5%. Although this variability is well recognized, few models exist to estimate HCC risk in patients with cirrhosis and none are commonly used. Liver societies recommend the same screening strategy (abdominal ultrasonography every 6 months with or without concomitant serum alpha-fetoprotein [AFP]) irrespective of HCC risk.<sup>1–3</sup> Studies show poor compliance with these screening recommendations.<sup>4,5</sup> Stratification of HCC risk in patients with cirrhosis into low (*e.g.* <1% per year), medium (*e.g.* 1–3% per year) and high (*e.g.* >3% per year) would enable optimization and individualization of outreach efforts and screening strategies in patients with cirrhosis. It would also enable identification of high-risk patients for clinical trials of HCC screening.

Many determinants of HCC risk in cirrhotic patients have already been described including routinely available clinical characteristics (*e.g.* etiology of cirrhosis, age, gender, race, body mass index, diabetes, HCV genotype) and serum laboratory tests (*e.g.* platelet count, albumin, aspartate aminotransferase [AST], alanine aminotransferase [ALT] and AFP).<sup>6</sup> These readily available predictors could potentially be combined into HCC risk estimation models with adequate discrimination and calibration.

The incidence of HCC has been rising dramatically over time.<sup>7,8</sup> Therefore, HCC risk prediction models must be based on a recent cohort to avoid underestimating HCC risk. HCC risk is frequently calculated in patients with cirrhosis from the time they were first diagnosed with cirrhosis (*i.e.* as an "inception cohort"). However, this results in the diagnosis of many prevalent cases as well as incident cases in the early period after the





Keywords: Liver cancer; Cirrhosis; Risk model; Surveillance; HCC; NAFLD; Non-alcoholic fatty liver disease.

Received 27 October 2018; received in revised form 12 March 2019; accepted 3 May 2019; available online 28 May 2019

<sup>\*</sup> Corresponding author. Address: Veterans Affairs Puget Sound Health Care System Gastroenterology, S-111-Gastro, 1660 S. Columbian Way, Seattle, WA 98108, United States; Tel.: +1-206-277-3136, fax +1-206-764-2232.

E-mail addresses: georgei@medicine.washington.edu, @gnioannou (G.N. Ioannou).

inception of the cohort. For practicing clinicians, a more useful scenario would be to estimate HCC risk in a patient with cirrhosis when they are seen in clinic at some point during their natural history, not necessarily only when they first get diagnosed with cirrhosis. Also, HCC risk prediction models developed specifically to inform screening strategies, need to have a time horizon of approximately 5 years or less. This is counterintuitive since longer follow-up is usually considered better; however, HCCs destined to develop in 5 to 10 years from now are not going to be diagnosed by screening occurring "now" and therefore should not influence the prediction models.

The 3 most common etiologies of cirrhosis in the United States, which also account for the majority of HCC cases, are hepatitis C virus (HCV), alcoholic liver disease (ALD) and nonalcoholic fatty liver disease (NAFLD). HCC risk prediction models need to be developed separately for different major etiologies of cirrhosis for a number of reasons. Firstly, HCC risk is greater in cirrhotic patients with HCV than those with other underlying liver diseases.<sup>6,9</sup> Secondly, certain predictors may be more important for some etiologies than others (e.g. obesity and diabetes may be more important in NAFLD-cirrhosis than in HCVcirrhosis) while some predictors are unique to some etiologies (e.g. HCV genotype is unique to HCV-cirrhosis). Thirdly, directacting antiviral treatments have a dramatic effect on HCC risk in HCV-related cirrhosis, such that HCC risk has to be calculated specifically relative to the time of receipt of antiviral treatment. The prevalence of HCV-related cirrhosis is declining rapidly due to the decreasing prevalence of HCV since the year 2001 and the more recent dramatic increase in HCV cures after the introduction of direct-acting antiviral treatments. In contrast, the prevalence of cirrhosis related to NAFLD and ALD is increasing.<sup>7,10</sup>

For these reasons, we aimed to develop and internally validate models that accurately estimate HCC risk in a recent cohort of patients with cirrhosis, specifically in patients with NAFLDcirrhosis or ALD-cirrhosis. Furthermore, we wanted to evaluate the net benefit that would be derived by implementing HCC surveillance strategies based on HCC risk estimated by our models. Finally, we wanted to make our HCC risk prediction models available to clinicians as web-based tools so that HCC risk can be readily estimated in clinical practice.

# Patients and methods

## Study design

We identified all patients with ALD-cirrhosis or NAFLD-cirrhosis who received care in the Veterans Affairs (VA) healthcare system (VAHS) nationally in calendar year 2012, without a prior history of HCC. We followed them retrospectively from 2012 until 01/01/2018 for the development of incident HCC. We used baseline characteristics in 2012 to develop and internally validate HCC risk prediction models separately for ALD-cirrhosis and NAFLD-cirrhosis.

## Study population and data source

The VAHS is the largest integrated healthcare system in the United States, with 168 VA Medical Centers and 1,053 outpatient clinics throughout the United States serving more than 8.9 million Veterans each year,<sup>11</sup> including the largest number of patients with cirrhosis.<sup>7</sup> Nationwide, the VA uses a single comprehensive electronic healthcare information network which integrates all care applications into a single, common database. We derived electronic data on all VA patients with cirrhosis using the VA Corporate Data Warehouse (CDW), a national, continually updated repository of electronic VA data developed specifically to facilitate research.<sup>12</sup> Data extracted included all patient pharmacy prescriptions, demographics, inpatient and outpatient visits, problem lists, procedures, vital signs, diagnostic tests, and laboratory tests. The study was approved by the Institutional Review Board of the Veterans Affairs Puget Sound Healthcare System and a waiver of informed consent was granted.

We identified 62,030 patients who had a diagnosis of cirrhosis first recorded at or before 12/31/2012 and who received VA healthcare in calendar year 2012, defined by having at least 1 inpatient or outpatient visit for any indication during that calendar year. We excluded 2,456 patients who had undergone liver transplantation before the beginning of follow-up, 2,970 patients who had a diagnosis of HCC first recorded before the beginning of follow-up, and 3,933 patients without available laboratory data in 2012. Among the remaining 52,671 patients, 16,175 had ALD-cirrhosis and 7,068 had NAFLD-cirrhosis (defined below) and were included in this study, while the remaining 29,428 had other etiologies of cirrhosis (mostly HCV) and were excluded.

## **Definition of cirrhosis**

The diagnosis of cirrhosis was based on the presence of the ICD-9 codes for cirrhosis or complications of cirrhosis listed in Table S1, recorded at least twice in any inpatient or outpatient encounter. This approach has been validated and widely used in VA-based studies by  $us^{6.7,13-19}$  and others.<sup>20-24</sup> The diagnosis of cirrhosis using *a single* ICD-9 code in VA data has been shown to have a 90% positive predictive value (probability that cirrhosis is present among those with a code) compared to chart extraction.<sup>25</sup> By requiring the relevant ICD-9 codes to be recorded at least twice we have found that the positive predictive value increased to 97% in a random sample of 250 patients included in the current study.

Among patients with cirrhosis, we defined ALD-cirrhosis and NAFLD-cirrhosis based on our previously published studies<sup>7</sup> as follows:

<u>ALD-cirrhosis:</u> Patients with ICD-9 codes for alcohol use disorders (Table S1) in the absence of serologic or virologic markers of chronic HCV or hepatitis B virus (HBV) infection and in the absence of ICD9 codes for hemochromatosis, primary biliary cirrhosis, primary sclerosing cholangitis, and autoimmune hepatitis (Table S1).

<u>NAFLD-cirrhosis</u>: was defined in patients with diabetes (ICD-9 code 250–250.92, recorded at least twice<sup>25</sup>) or body mass index (BMI)  $\geq$  30 kg/m<sup>2</sup> prior to the diagnosis of cirrhosis, who did not have alcohol use disorders, serologic or virologic markers of chronic HCV or HBV infection or ICD9 codes for hemochromatosis, primary biliary cirrhosis, primary sclerosing cholangitis, and autoimmune hepatitis (Table S1). NAFLD-related cirrhosis does not have pathognomonic serological, radiological, or histological features – even hepatic steatosis is frequently absent after cirrhosis develops. Hence we adapted a clinical definition of NAFLD based on previous work<sup>7,26</sup> that reflects the diagnostic process used in clinical practice, in which NAFLD is suspected in the presence of risk factors such as obesity and diabetes after exclusion of other etiologies.

Table 1. Baseline characteristics of our cohort of patients with cirrhosis at the time of cohort inception in 2012, presented according to cirrhosis etiology.

		Etiology o	Etiology of cirrhosis			
	ALD or	ALD	NAFLD	p value		
	NAFLD	n = 16,175	n = 7068			
	n = 23,243					
Age, yr (mean ± SD)	$64.2 \pm 9.4$	$62.9 \pm 8.9$	67.1 ± 9.7	< 0.001		
Male (%)	97.2	97.9	95.5	< 0.001		
Race/ethnicity (%)				< 0.001		
White, non-Hispanic	73.4	72	76.4			
Black, non-Hispanic	10.1	11.1	7.5			
Hispanic	8	8.4	7.1			
Other	2.2	2.3	2			
Declined to answer/missing	6.4	6.1	6.9			
BMI, kg/m <sup>2</sup> (mean ± SD)	$29.9 \pm 6.7$	$28.5 \pm 6.3$	$33.0 \pm 6.6$	< 0.001		
AUDIT-C score <sup>†</sup> (%)						
No alcohol use	57.5	47.9	73.9			
Low-level alcohol use	20.1	19.3	19.8			
Unhealthy alcohol use	22.2	29.4	3.0	< 0.001		
Diabetes (%)	48.4	35.5	77.9	< 0.001		
Substance use disorder (%)	16.4	22.5	2.6	< 0.001		
HIV coinfection (%)	0.5	0.4	0.7	< 0.01		
Complications of cirrhosis (%)	37.7	39.3	34.2	< 0.001		
Ascites	19.8	22.2	14.4	< 0.001		
Encephalopathy	6.9	8.2	3.8	< 0.001		
Gastroesophageal varices (with bleeding)	5.4	6.1	4	< 0.001		
Gastroesophageal varices (without bleeding)	18.3	18.6	17.7	0.1		
Laboratory results, (mean ± SD)						
MELD score	11.2 ± 5.3	$11.0 \pm 5.2$	11.7 ± 5.7	< 0.001		
AFP <sup>*</sup> (ng/ml)	$3.3 \pm 2.6$	$3.4 \pm 2.6$	$3.0 \pm 2.4$	0.35		
Hemoglobin (g/dl)	13.1 ± 2.3	13.2 ± 2.3	13.0 ± 2.2	< 0.001		
Platelet count $(k/\mu l)$	158 ± 83	162 ± 86	150 ± 76	< 0.001		
Creatinine (mg/dl)	$1.1 \pm 0.8$	$1.0 \pm 0.6$	$1.3 \pm 1.0$	< 0.001		
Bilirubin (g/dl)	$1.4 \pm 2.2$	$1.5 \pm 2.5$	$1.0 \pm 1.2$	< 0.001		
INR	$1.4 \pm 1.2$	1.3 ± 1.1	1.4 ± 1.3	< 0.001		
Albumin (g/dl)	$3.6 \pm 0.7$	$3.6 \pm 0.7$	$3.7 \pm 0.6$	< 0.001		
Alkaline phosphatase (U/L)	113 ± 79	116 ± 82	$105 \pm 70$	< 0.001		
AST/ <sub>V</sub> /ALT ratio	$7.8 \pm 5.1$	$8.4 \pm 5.6$	6.5 ± 3.1	< 0.001		
FIB-4 score	5.1 ± 23.0	5.3 ± 24.1	$4.6 \pm 20.4$	0.03		

AFP, alpha-fetoprotein; ALD, alcohol-related liver disease; ALT, alanine aminotransferase; AST, aspartate aminotransferase; AUDIT-C, Alcohol Use Disorders Identification Test Consumption; BMI, body mass index; FIB-4, fibrosis-4; INR, international normalized ratio; MELD, model for end-stage liver disease; NAFLD, non-alcoholic fatty liver disease.

\* Median and interquartile range shown for AFP due to substantial left skewing.

<sup>†</sup> No alcohol use: AUDIT-C score 0. Low-level alcohol use: AUDIT-C 1–3 in men, 1–2 in women. Unhealthy alcohol use: AUDIT-C 4–12 in men, 3–12 in women.

## **Baseline patient characteristics**

We ascertained the values of all baseline characteristics as of calendar year 2012 (the year of inception of our cohort). We extracted age, sex, race/ethnicity and all the baseline laboratory tests shown in Table 1. BMI was calculated as the measured weight in kilograms divided by the square of the measured height in meters. For patients with multiple laboratory tests or BMI measurements we recorded the one closest to 01/01/2012. We also determined the presence of complications of cirrhosis (ascites, spontaneous bacterial peritonitis, encephalopathy, gastroesophageal varices with or without bleeding, hepatorenal syndrome, hepatopulmonary syndrome), type 2 diabetes mellitus, alcohol use disorders, substance use disorders and HIV infection based on appropriate ICD-9 codes recorded at least twice prior to the date of entry into the cohort in 2012 (Table S1). These ICD9-based definitions of cirrhosis, decompensated cirrhosis and other comorbidities have been widely used and validated in studies using VA medical records.<sup>7,19–24</sup> Since 2004, the VA has implemented annual screening for alcohol use disorders using the validated Alcohol Use Disorders Identification Test Consumption (AUDIT-C). We extracted the AUDIT-C score within 12 months prior to entry into the cohort.

#### Diagnosis of hepatocellular carcinoma

The diagnosis of HCC was based on the presence of ICD-9 code 155.0 and ICD-10 code C22.0 (the VA switched to ICD-10 codes on 10/1/2015) recorded at least twice. The ICD-9 code-based definition of HCC using VA records has been shown to have a positive predictive value of 84–94% compared to chart extraction<sup>23,27–28</sup> and has been widely used by us<sup>6–7,18,29</sup> and other investigators.<sup>30–32</sup>

## Statistical analysis

We used multivariable Cox proportional hazards regression to model the risk of HCC using baseline characteristics ascertained in 2012 separately in patients with cirrhosis related to ALD or NAFLD. Follow-up started in 2012 (specifically the date of the latest blood draw during the period 1/1/2011 to 12/31/2012) and continued until 01/01/2018. We did not exclude any period of time following that blood draw from analysis (*e.g.* 3 or 6 months), even recognizing that there might be a higher

likelihood of prevalent cases at the beginning, because we wanted the models to reflect the clinical scenario of a patient being seen by their provider in clinic. Patients who did not develop HCC by 01/01/2018, were censored at the time of death, liver transplantation or the date of last follow-up in the VA. We contemplated reporting a competing risks analysis, accounting for death as a competing risk, but it made little difference to the models and would have unnecessarily complicated the methodology.

## **Model building**

We considered 25 characteristics listed in Table 1 as potential predictors of HCC for inclusion in our models. We used an iterative process to determine which predictors to include in our final models. First, we estimated measures of discrimination, calibration, and significance when each predictor was added to the base model and identified the top 5 predictors with the greatest improvement in these measures. We chose the predictor that was consistently in the top 5 with a preference for a *p* value <0.10 and an improvement in the main measures of discrimination and calibration, namely Gönen and Heller's ĸ, Royston and Sauerbrei's D-statistic, calibration slope, and integrated Brier score (IBS). We verified graphically that the added predictor improved the observed versus predicted risk plot, thus allowing assessment over the entire time period. A pooled kfold cross-validation was used to calculate all measures of model performance. A k equal to 10 was chosen to address the bias versus variability in a database with a large sample size, but relatively few events.

We then updated the base model to include the chosen predictor and removed any predictors with a p value <0.10; removed predictors were added back into the list of potential predictors. We favored variables that are objectively ascertained (such as laboratory test) and those that have been consistently associated with HCC in previous studies (such as age and sex).

We considered both dummy-categorical as well as continuous (linear or transformed) modeling of laboratory tests. Interaction terms were explored in cases where there were biological indications. The distribution of the model predictions was checked for normality.

Measures of model discrimination, calibration and accuracy Discrimination refers to the ability to separate those who will get HCC from those who will not. The measures of discrimination chosen were Harrell's C-index<sup>33</sup> (which measures the degree of concordance between pairs and is sensitive to censoring), Gönen and Heller's  $\kappa^{34}$  (which is also a measure of concordance but is robust to censoring), and Royston and Sauerbrei's D-statistic<sup>35</sup> (which can be interpreted as the log hazard ratio of risk between the low and high risk groups dichotomized at their median values). Calibration refers to the degree of agreement between model-derived probabilities and observed probabilities. As a calibration measure, we evaluated the calibration slope,<sup>36</sup> which is robust to censoring and ideally takes a value of 1. Additionally, we evaluated calibration graphically, by comparing the observed Kaplan-Meier estimates of HCC-free survival and lowess-smoothed model predictions of HCC-free survival after categorizing risk into low, medium, or high groups. Model prediction accuracy was evaluated using the IBS,<sup>37</sup> which is the mean squared difference between the predicted probability and the actual outcome and the area under the receiver operating characteristic curve (AUROC), derived from a logistic regression of model predictions and diagnosis of HCC.

Measures of discrimination and calibration were estimated using k-fold cross-validation during model building. In addition, to obtain the most conservative estimates of discrimination and calibration, we split the data in half, into derivation and validation datasets balanced on number of events and on VA facility (*i.e.* patients from approximately half of the VA facilities were included in the derivation dataset and from the other half in the validation dataset).<sup>38</sup> Measures of assessment were then calculated for each dataset using model coefficients from the derivation data.

We also calculated the performance measures of the "Toronto HCC risk index" in predicting HCC risk in our study population as a comparison.<sup>39</sup> This is a recently published HCC risk prediction tool that uses 4 predictors: age, sex, etiology of liver disease, and platelet count. Additionally, for comparison, we calculated the performance measures of the fibrosis-4 (FIB-4) score,<sup>40</sup> which incorporates age, platelet count, and AST/ $\sqrt{ALT}$ , and correlates with HCC, and the model for end-stage liver disease (MELD) score, as a measure of liver dysfunction.

# Use of decision curves to estimate the net benefit of using our risk prediction models

We used decision curves to estimate the standardized net benefit that would be expected in a population if our models are used to estimate HCC risk and patients are screened only when their estimated risk exceeds a pre-established risk threshold, compared to the "screen-all" approach. A decision curve is a novel graphical plot of net benefit versus risk threshold that was proposed in 2006 for assessing the potential population impact of adopting a risk prediction instrument.<sup>41</sup> A risk threshold is defined as that probability of HCC above which screening would be favorable over not screening. "Standardized" net benefit is the proportion of total possible net benefit, which would be achieved by a "perfect" risk model that places all the patients with HCC above the risk threshold for screening without any false positives (*i.e.* without placing patients without HCC above the risk threshold). To avoid over-fitting, decision curves were calculated using repeated 10-fold crossvalidation<sup>41</sup> The cross-validation was repeated 50 times and results were averaged.

## **Results**

### Characteristics of study population

Among 23,243 patients with ALD-cirrhosis or NAFLD-cirrhosis in VA care in 2012, most were male (97%) and non-Hispanic White (73%) (Table 1). Compared to patients with ALDcirrhosis, those with NAFLD-cirrhosis were older (67.1 vs. 63.0 years old), had higher BMI (33.0 vs. 28.5 Kg/m<sup>2</sup>) and were more likely to have diabetes (78% vs. 36%). Substance use disorders were more common in ALD-cirrhosis than NAFLDcirrhosis (22.5% vs. 2.6%). By definition, alcohol use disorders were universal in ALD-cirrhosis and absent in NAFLDcirrhosis. Only 3% of patients in the NAFLD-cirrhosis group reported unhealthy alcohol use (AUDIT-C 4-12 in men or 3-12 in women) - which further confirmed our diagnostic definition of NAFLD-cirrhosis -compared to 29.4% in the ALDcirrhosis group. Complications of portal hypertension were present in a similar proportion in both groups at baseline (37.7%) and mean MELD score was 11.2%.



Fig. 1. Cumulative incidence curves showing the probability of developing HCC in a cohort of patients followed from 2012 to 2018. Plotted by (A) etiology of cirrhosis (ALD vs. NAFLD) and (B) FIB-4 score >3.25 or  $\leq$ 3.25. (C) Table showing the incidence of HCC in this population by cirrhosis etiology and FIB-4 score. ALD, alcohol-related liver disease; FIB-4, fibrosis-4; HCC, hepatocellular carcinoma; NAFLD, non-alcoholic fatty liver disease.

#### **HCC incidence**

During a mean follow-up of 3.7 years (range 1 to 6), 1,278 out of 23,243 patients with cirrhosis developed HCC (Fig. 1). The annualized incidence of HCC was similar in ALD-cirrhosis (1.44%) and NAFLD-cirrhosis (1.56%) and mean follow-up practically identical (3.7 years). The annual incidence was significantly greater in the subset of patients with FIB-4 >3.25 (2.68%) than FIB-4 <3.25 (0.68%). However, a substantial proportion of HCCs (354/1,278 or 28%) occurred in patients with FIB-4 <3.25, justifying our decision not to exclude patients with a diagnosis of cirrhosis who had a FIB-4 <3.25.

## **HCC predictors**

Older age, male sex and low platelet count were strong, independent predictors of HCC, consistent with many published studies (Table 2). In addition, low albumin and high  $AST/\sqrt{ALT}$ 

## JOURNAL OF HEPATOLOGY

were strong, independent predictors of HCC. Increasing BMI was a significant predictor in ALD-cirrhosis but not NAFLDcirrhosis and diabetes was a stronger predictor in ALDcirrhosis than NAFLD-cirrhosis. However, the associations of BMI and diabetes with HCC in patients with NAFLD-cirrhosis were blunted because presence of diabetes or BMI >30 were used as defining characteristics for NAFLD-cirrhosis.

#### **Development of models estimating HCC risk**

Out of the 25 potential predictors that we considered (shown in Table 1), 7 were included in the final models that we developed: age, sex, BMI, diabetes, platelet count, serum albumin and serum AST/ $\sqrt{ALT}$  ratio (Table 2). The same 7 predictors were selected for the models in NAFLD-cirrhosis and ALD-cirrhosis. However, the coefficients (and hazard ratios) for each predictor were slightly different for each, especially for diabetes and BMI. Of these 7 predictors, 4 predictors (age, platelet count, serum AST/ $\sqrt{ALT}$  ratio and albumin) accounted for most of the prediction, *i.e.* accounted for most of the explained risk<sup>42</sup> (93.9% in NAFLD-cirrhosis and 94.0% in ALD-cirrhosis). The coefficients of the models are shown in Table S2 and the baseline cumulative incidences at years 1–5 in Table S3, providing all the necessary information for other investigators to apply or externally validate the models.

We also developed a model for the combined population of NAFLD-cirrhosis and ALD-cirrhosis (Table 2), which included the same 7 predictors. We considered including the type of cirrhosis as a covariate in the model, but it was not a significant predictor and hence was omitted.

Measures of discrimination were generally very good and slightly higher for the ALD-cirrhosis than the NAFLD-cirrhosis model (Table 3). For example, the Harrell's C index in the validation half of the dataset was 0.74 for ALD-cirrhosis and 0.72 for NAFLD-cirrhosis. Predicted versus observed cumulative incidence curves showed great overlap (Fig. 2). The calibration slope was excellent for all models as was the prediction accuracy measured by the IBS. The AUROC in the validation-half of the dataset was ~0.75, which is considered very good.

Using the published "Toronto HCC Risk Index"<sup>39</sup> to estimate HCC risk resulted in lower Harrell's C index (0.67) and AUROC (0.68) than when using our models. The FIB-4 score, which is calculated as the product of 3 measurements (age, inverse platelet count and AST/ $\sqrt{ALT}$ ) all of which are strong predictors of HCC, performed well with a Harrell's C (0.70) and AUROC (0.71) that were slightly worse than our models. The MELD score performed poorly (Harrell's C 0.598, AUROC 0.527).

# Standardized net benefit of model-based HCC surveillance ascertained by decision curves

Decision curves plotted in Fig. 3 show that the risk model-based screening strategy has superior standardized net benefit than the "screen-all" strategy if the screening threshold is >1.5% per year (or >7.5% over 5 years) for ultrasound-based screening, as recommended by the American Association for the Study of Liver Diseases (AASLD) guidelines<sup>43</sup>. The decision curves also show that screening based on risk estimates derived from our model (*i.e.* screening patients who exceed a given risk threshold) is superior to the screen-all strategy for a wide range of plausible screening thresholds. For more intensive screening strategies, such as those using abbreviated MRI, which are proposed for patients at even higher HCC risk (*e.g.* >3% per year), the risk-based screening is superior to "screen-all".

Table 2. Models developed to estimate HCC risk in	patients with ALD-Cirrhosis, NAFLD-cirrhosis or the combined pop	ulation.
Tuble 2. mouth atteroped to totimate mot	putternes then the entities of the combined pop	

Predictors <sup>†</sup>	ors <sup>†</sup> ALD-cirrhosi		NAFLD	-cirrhosis	ALD or NAFL	ALD or NAFLD cirrhosis		
	aHR	p value	aHR	p value	aHR	p value		
Age, yr								
≤49	1		1		1			
>49-59	1.91	< 0.01	1.39	0.42	1.83	<0.01		
>59-64	2.84	< 0.001	2.01	0.08	2.69	< 0.001		
>64-69	3.16	< 0.001	2.37	0.03	3.05	< 0.001		
>69-77	3.76	< 0.001	1.95	0.1	3.15	< 0.001		
>77	3.4	< 0.001	2.09	0.07	3.05	< 0.001		
Sex								
Male	1		1		1			
Female	0.6	0.16	0.24	< 0.01	0.39	< 0.01		
BMI, kg/m <sup>2</sup>								
≤25.2	1		1		1			
>25.2-29.3	1.32	< 0.01	0.91	0.64	1.25	0.01		
>29.3-33.8	1.39	< 0.01	0.9	0.6	1.29	< 0.01		
>33.8-38.5	1.49	< 0.001	0.8	0.28	1.27	0.02		
>38.5	1.31	0.06	0.77	0.22	1.15	0.22		
Diabetes								
No	1		1		1			
Yes	1.46	< 0.001	1.24	0.1	1.39	< 0.001		
Platelet count, k/µl								
>203	1		1		1			
>146-203	1.89	< 0.001	1.12	0.61	1.64	< 0.001		
>99-146	2.64	< 0.001	2.18	<0.001	2.51	< 0.001		
>68-99	2.99	< 0.001	3.04	<0.001	3.04	< 0.001		
≤68	4.13	< 0.001	3.52	< 0.001	3.92	< 0.001		
Albumin, g/dl								
>4.1	1		1		1			
>3.7-4.1	1.4	< 0.01	1.49	0.03	1.43	< 0.001		
>3.3-3.7	2.2	<0.001	2.11	< 0.001	2.18	< 0.001		
>2.8-3.3	2.6	< 0.001	2.73	< 0.001	2.66	< 0.001		
≤2.8	3.5	< 0.001	3.41	< 0.001	3.48	< 0.001		
AST/ <sub>1</sub> /ALT								
≤5.02	1		1		1			
>5.02-6.45	1.63	<0.001	1.27	0.13	1.47	< 0.001		
>6.45-8.80	2.69	< 0.001	1.99	< 0.001	2.4	< 0.001		
>8.80-12.83	3.27	< 0.001	2.5	< 0.001	2.95	< 0.001		
>12.83	2.71	< 0.001	4.99	< 0.001	2.67	< 0.001		

The table shows the adjusted hazard ratios (and their *p* values) for each predictor included in the models. The models are available to be executed at www.hccrisk.com. Coefficients and baseline cumulative incidences are reported in Tables S2 and S3.

aHR, adjusted hazard ratio; ALD, alcohol-related liver disease; ALT, alanine aminotransferase; AST, aspartate aminotransferase; BMI, body mass index; HCC, hepatocellular carcinoma; NAFLD, non-alcoholic fatty liver disease.

<sup>†</sup> The continuous variables (BMI, platelet count, albumin, and AST/ $\sqrt{ALT}$ ) were categorized according to the 0–25th, 25–50th, 50–75th, 75–90th, and >90th percentiles.

## Web-based HCC risk estimating tools (available at www. hccrisk.com)

#### Discussion

We implemented the 2 models shown in Table 2 for ALDcirrhosis and NAFLD-cirrhosis as web-based tools to allow clinicians to estimate HCC risk in individual patients. The models can be executed at www.hccrisk.com. Table 4 shows 5-year HCC risk estimated by our models using the baseline characteristics shown in selected patients in "low-risk" (annualized risk 0-1%), "medium-risk" (annualized risk >1-3%) and "high-risk" (annualized risk >3%) categories. Table 4 shows that 5-year risk can vary from as low as 0.4% to as high as 30% depending on the values of the 7 simple predictors included in our models. We envision that the web-based HCC risk calculator will be used in the future to risk-stratify patients into low, medium and high-risk categories for the purposes of outreach efforts to improve screening uptake, development of future risk-based screening strategies and selection of patients for clinical trials of screening (Fig. 4).

We demonstrated that HCC risk (i.e. incidence) varies dramatically in patients with ALD-cirrhosis or NAFLD-cirrhosis, which questions the utility of having a single screening strategy for all patients. We developed and internally validated models that accurately estimate HCC risk in patients with NAFLD-cirrhosis or ALD-cirrhosis according to 7 readily available characteristics: age, sex, BMI, diabetes, platelet count, serum albumin and serum AST/v/ALT ratio. HCC surveillance strategies based on HCC risk estimates derived from our models resulted in greater predicted net benefit that the current "one-size-fits-all" strategy. Our models can be used to stratify patients according to estimated annual HCC risk into low-risk (e.g. annual risk <1%), medium-risk (e.g. annual risk 1-3%) and high-risk (e.g. annual risk >3%). Such risk stratification can be used to select patients for clinical trials, to inform screening outreach efforts and to develop risk-based HCC screening strategies, such as more intensive screening strategies (e.g. abbreviated MRI) for

		Discrimination	Calibration	Accuracy		
	Gonen and Heller's κ-statistic	Royston and Sauerbrei's D-statistic	Harrell's C	Calibration slope	Integrated Brier score	AUROC
ALD-cirrhosis						
k-fold cross-validation	0.737	1.452	0.743	0.947	0.04	0.750
Derivation-half	0.741	1.546	0.755	1	0.034	0.764
Validation-half	0.740	1.431	0.740	0.913	0.047	0.740
NAFLD-cirrhosis						
k-fold cross-validation	0.730	1.379	0.720	0.888	0.037	0.739
Derivation-half	0.739	1.695	0.749	1	0.033	0.775
Validation-half	0.736	1.236	0.718	0.74	0.040	0.721
ALD- or NAFLD-cirrhosis						
k-fold cross-validation	0.733	1.436	0.739	0.957	0.039	0.749
Derivation-half	0.742	1.58	0.757	1	0.033	0.761
Validation-half	0.742	1.335	0.727	0.845	0.046	0.738
Toronto HCC Risk Score <sup>39</sup>			0.669			0.679
FIB-4 score			0.701			0.713
MELD score			0.598			0.527

Gonen and Heller's  $\kappa$ -statistic is a concordance measure and a value of 1 indicates perfect discrimination. <u>Royston and Sauerbrei's</u> D-statistic is a hazard ratio and the greater than 1 the greater the discrimination. <u>Harrell's C</u> statistic is a concordance measure and a value of 1 indicates perfect discrimination. A <u>Calibration slope</u> of 1 indicates perfect calibration. An <u>Integrated Brier score</u> of 0 indicates perfect accuracy. It is the mean squared difference between the predicted probability and the actual outcome. <u>AUROC:</u> A value of 1 indicates perfect accuracy.

ALD, alcohol-related liver disease; AUROC, area under the receiver operating characteristic curve; FIB-4, fibrosis-4; HCC, hepatocellular carcinoma; MELD, model for endstage liver disease; NAFLD, non-alcoholic fatty liver disease.

The measures are shown separately for the derivation and validation datasets.

high-risk patients. Future research should evaluate specific riskbased strategies, such as screening by abbreviated MRI versus ultrasonography for patients in the high-risk category. Our models are available as web-based tools that can be used to estimate HCC risk in individual patients (www.hccrisk.com).

HCC risk varies widely in cirrhotic patients.<sup>6</sup> In the patients depicted in Table 4, estimated HCC risk varied by more than 30-fold depending on the distribution of readily available HCC risk factors. Despite this, national guidelines recommend a "one-size-fits-all" surveillance strategy, whereby the same screening strategy (biannual ultrasonography ± AFP) is recommended for all patients with cirrhosis<sup>1</sup> irrespective of HCC risk. We believe that great improvements in HCC screening can be achieved if patients are first stratified according to HCC risk and then offered risk-appropriate screening (Fig. 4). However, no widely accepted method is available for estimating HCC risk in cirrhotic patients. Accurate estimation of HCC risk by the models that we developed could potentially improve HCC surveillance efforts, increase early detection of HCC, reduce harms related to unnecessary surveillance and facilitate the design of future HCC surveillance trials. Patients at high risk of HCC could be targeted for interventions to improve their uptake of HCC surveillance. It is currently estimated that ≤20% of cirrhotic patients undergo surveillance consistent with guidelines in the United States.<sup>44,45</sup> Different surveillance strategies could potentially be proposed for different categories of HCC risk. For example, more effective strategies that are also more expensive or more invasive/harmful, such as screening by MRI, abbreviated MRI or CT, would be more cost-effective if they focus on the higher risk groups.<sup>46</sup> In healthcare systems with limited resources unable to support universal surveillance of all cirrhotic patients, surveillance could be limited to patients with higher HCC risk. Past AASLD guidelines recommended that "for patients with cirrhosis, surveillance should be offered when the risk of HCC is >1.5% per year".<sup>43</sup> If the HCC risk could be accurately predicted, those with incidence well below 1.5% may not be recommended surveillance thus reducing costs and potential downstream harms such as unnecessary CT scans (radiation and iv contrast), liver biopsies and other procedures (assuming that the risk threshold of >1.5% is indeed valid).<sup>47</sup> Estimation of HCC risk could enable individualized counseling of patients by their providers leading to improved compliance with surveillance recommendations and engagement in care. Finally, our models could be used to identify high-risk patients for participation in clinical trials of HCC surveillance or HCC risk modifiers.

We hope that the strategy of HCC risk stratification will be utilized more widely in patients with cirrhosis by individual physicians for individual patients or by healthcare systems for patient populations. To that end we created web-based tools to execute our models and estimate HCC risk in patients with NAFLD-cirrhosis or ALD-cirrhosis available at www.hccrisk.com. We also provided all the coefficients and baseline cumulative incidence rates of the models (Tables S2 and S3) to allow other investigators to execute or externally validate the models. These new models extend the models we published recently estimating HCC risk in patients with HCV infection,<sup>48</sup> also available on the same website.

Notable efforts have been made recently to develop models estimating HCC risk in patients with cirrhosis using readily available predictors.<sup>39,49</sup> The best example is probably the "Toronto HCC risk index", which combined 4 predictors (age, sex, etiology of liver disease, and platelet count) into an HCC risk prediction model that performed reasonably well and was externally validated.<sup>39</sup> However, the majority of the patients in that study (1,279 out of 2,079) had viral hepatitis B and C, in whom antiviral treatment dramatically reduces HCC risk, a factor not captured in the prediction model. We believe that distinct models need to be developed for cirrhotic patients with cured HCV<sup>48</sup> and adequately treated HBV, as well as for patients



**Fig. 2. Predicted versus observed cumulative incidence of HCC based on predictive models.** Developed for (A) ALD-cirrhosis; (B) NAFLD-cirrhosis and (C) ALD or NAFLD-cirrhosis. Patients in each subgroup are divided into thirds (low, medium and high) based on the predicted risk. The plots show excellent overlap between observed and predicted cumulative incidence. ALD, alcohol-related liver disease; HCC, hepatocellular carcinoma; NAFLD, non-alcoholic fatty liver disease.

with NAFLD and ALD-cirrhosis, as we have done. Indeed, when we used the "Toronto HCC risk index" in our study population we obtained a lower Harrell's C index (0.67) than that reported in their study population (0.76). The "Adress-HCC" model was developed among patients on liver transplant waitlists and used 6 baseline clinical variables to estimate HCC risk (age, diabetes, race, etiology of cirrhosis, sex, and severity of liver dysfunction).<sup>49</sup> However, median follow-up was short (1.3 years) and patients were limited to those waitlisted for transplantation. The field of HCC risk calculation is still in its infancy and it is likely the models will continue to evolve, possibly incorporating



Fig. 3. Decision curves comparing the standardized net benefit achieved by screening based on HCC risk. Predicted by the model (i.e. screening only patients who exceed a certain threshold probability - blue line) to the 'screen-all" (green line) or "screen-none" (orange line) strategies. The vertical axis shows standardized net benefit, which is the proportion of total possible net benefit and would be achieved by a risk model with 100% sensitivity and specificity. The horizontal axis shows different 5-year HCC risk thresholds that might be used to recommend screening. For example, the American Association for the Study of Liver Diseases recommends screening when annual HCC risk exceeds 1.5% in patients with cirrhosis, or 5-year risk exceeds 7.5%, marked with a vertical dotted line. The Figures illustrate that the net benefit of screening based on our models (blue line) is greater than the net benefit of the "screen-all" strategy (green line), for both patient groups. The figures also show that for a wide range of plausible screening thresholds the risk model-based screening has superior standardized net benefit than the screen-all strategy. ALD, alcohol-related liver disease; HCC, hepatocellular carcinoma; NAFLD, non-alcoholic fatty liver disease.

Table 4. Estimates of 5-year HCC risk calculated by our web-based models for selected patients in the "low-risk", "medium-risk" and "high-risk" categories.

	Low-risk patients 5-year HCC risk < 5% Annual HCC risk < 1%			Medium-risk patients 5-year HCC risk 5% to 15% Annual HCC risk 1 to 3%			High-risk patients 5-year HCC risk > 15% Annual HCC risk > 3%					
Cirrhosis etiology	NAFLD	ALD	NAFLD	ALD	NAFLD	ALD	NAFLD	ALD	NAFLD	ALD	NAFLD	ALD
Age	61	67	62	62	66	60	62	66	63	62	63	62
Sex	F	М	М	М	F	М	М	М	М	М	М	М
Diabetes	No	No	Yes	No	Yes	Yes	No	Yes	Yes	Yes	Yes	Yes
BMI, kg/m <sup>2</sup>	31	28	31	28	35	23	35	24	37	29	37	29
Albumin, g/dl	4.2	4.2	3.8	3.9	3.6	3.9	3.5	3.1	3.8	3.4	3.4	3.3
Serum AST, IU/L	30	30	30	30	45	50	55	40	55	50	55	50
Serum ALT, IU/L	35	30	35	30	20	20	20	20	25	20	25	20
Platelet count, k/µl	225	170	170	120	85	110	115	180	90	110	85	50
Estimated 5-year HCC risk, %	0.40	2.6	3.5	4.6	6.41	10.0	12.8	14.4	15.5	20.6	21.3	30.2

ALD, alcohol-related liver disease; ALT, alanine aminotransferase; AST, aspartate aminotransferase; BMI, body mass index; HCC, hepatocellular carcinoma; NAFLD, non-alcoholic fatty liver disease.



**Fig. 4. HCC risk stratification using prediction models.** Risk estimation models can be used to stratify risk in patients. Outreach efforts to improve screening uptake, future screening strategies and clinical trials can utilize the categorization of patients into low, medium and high-risk categories. ALD, alcohol-related liver disease; HCC, hepatocellular carcinoma; NAFLD, non-alcoholic fatty liver disease.

novel biomarkers and likely developing different models for distinct subpopulations of patients with cirrhosis.

Decision curves plot the standardized net benefit that would be expected from risk-based screening strategies at different "appropriate" HCC risk thresholds for screening. Fig. 3 shows that at a threshold of >1.5% per year (or 7.5% per 5 years), which is commonly recommended in patients with cirrhosis<sup>50</sup>, the net benefit is greater with screening based on our models (i.e. screening only patients with estimated HCC risk >1.5% per year) compared with screening all patients. However, if the appropriate risk threshold is much lower (<0.8% per 5 years) then there is no difference between the screen-all and model-based screening strategies. It is important to emphasize that decision curves cannot be used to determine the appropriate HCC risk threshold at which screening is deemed to be beneficial. Instead, this threshold needs to be determined by estimating the harms of missing a case relative to the harms of unnecessarily screening a non-case. Decision analytic theory suggests that if the harms of missing a case are x-times greater than the harms of unnecessarily screening a non-case, then the appropriate threshold for screening is a risk exceeding 1/(x + 1).<sup>51</sup> Therefore, the greater the harms of missing a case (or the greater the benefits of diagnosing a case) the lower the risk threshold at which screening is beneficial. Conversely, the greater the harms of screening the higher the risk threshold. Decision curves have been used before to evaluate the value of predictive models for recommending CT screening for lung cancer,<sup>52</sup> but not for HCC screening.

A major limitation of our study is that we have not externally validated our models yet in non-VA populations. We suspect that our models may be a good predictors in other similar populations *i.e.* unselected, predominantly male patients with cirrhosis not limited to transplant centers or tertiary referral centers. However, other models such as the "Toronto HCC risk index"<sup>39</sup> or the "Adress-HCC"<sup>49</sup> model may be better predictors in patients at tertiary referral or transplant centers similar to the ones used for their model development. It will be critical to externally validate our models in non-VA populations and also ideally in populations undergoing routine HCC surveillance (although the frequency of surveillance does not affect the incidence (or risk) of HCC in a given population). It is possible that ultimately different models might be developed for different subpopulations of patients with cirrhosis, just as we have already developed different models for HCV-infected patients undergoing antiviral treatment.

Substantial strengths of this study include the large sample size and the large number of incident HCC cases. Additionally, this study was performed recently and incorporated an appropriate length of follow-up. Baseline characteristics necessary for modeling were available. All patients were derived from a single, national healthcare system with fairly uniform practices and guidelines across its facilities.

In conclusion, we developed and internally validated models estimating HCC risk in patients with ALD-cirrhosis and NAFLDcirrhosis. These models, which are available as web-based tools, can help stratify patients according to HCC risk and, consequently, help determine an appropriate screening strategy based on a patient's calculated risk. An intensive screening strategy targeting those who exceed a certain predetermined HCC risk may be more efficacious and cost-effective than the current "screen-all" or "screen-none" strategies which depend solely on cirrhosis status.

## **Financial support**

The study was funded by a NIH/NCI grant R01CA196692 and VA CSR&D grant I01CX001156 to GNI.

## **Conflict of interest**

The authors declare no conflicts of interest that pertain to this work.

Please refer to the accompanying ICMJE disclosure forms for further details.

#### **Authors' contributions**

All authors approved the final version of the manuscript. George loannou is the guarantor of this paper. George loannou: Study concept and design, acquisition of data, statistical analysis and interpretation of data, drafting of the manuscript, critical revision of the manuscript, obtained funding. Pamela Green: Acquisition of data, creation of analytic variables Kristin Berry: Study design, analysis of data, critical revision of manuscript. Kathleen F. Kerr: Interpretation, statistical analysis of decision curves.

## **Role of funding source**

The funding source played no role in study design, collection, analysis or interpretation of data.

## Disclaimer

The contents do not represent the views of the U.S. Department of Veterans Affairs or the United States Government.

## Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.jhep.2019.05.008.

#### References

- [1] Heimbach J, Kulik LM, Finn R, Sirlin CB, Abecassis M, Roberts LR, Zhu A, et al. Aasld guidelines for the treatment of hepatocellular carcinoma. Hepatology 2017.
- [2] European Association For The Study Of The Liver and European Organisation For Research Treatment Of Cancer (EASL-EORTC) clinical practice guidelines: management of hepatocellular carcinoma. J Hepatol 2012;56:908–943.
- [3] Omata M, Cheng AL, Kokudo N, Kudo M, Lee JM, Jia J, et al. Asia-Pacific clinical practice guidelines on the management of hepatocellular carcinoma: a 2017 update. Hepatol Int 2017;11:317–370.
- [4] Singal AG, Tiro J, Li X, Adams-Huet B, Chubak J. Hepatocellular carcinoma surveillance among patients with cirrhosis in a population-based integrated health care delivery system. J Clin Gastroenterol 2017;51:650–655.
- [5] Singal AG, Yopp A, Skinner SC, Packer M, Lee WM, Tiro JA. Utilization of hepatocellular carcinoma surveillance among American patients: a systematic review. J Gen Intern Med 2012;27:861–867.
- [6] Ioannou GN, Splan MF, Weiss NS, McDonald GB, Beretta L, Lee SP. Incidence and predictors of hepatocellular carcinoma in patients with cirrhosis. Clin Gastroenterol Hepatol 2007;5:938–945.
- [7] Beste LA, Leipertz SL, Green PK, Dominitz JA, Ross D, Ioannou GN. Trends in burden of cirrhosis and hepatocellular carcinoma by underlying liver disease in US veterans, 2001–2013. Gastroenterology 2015; 149:1471–1482, e5.
- [8] El-Serag HB. Epidemiology of viral hepatitis and hepatocellular carcinoma. Gastroenterology 2012;142:1264–1273, e1.
- [9] Ioannou GN, Green P, Lowy E, Mun EJ, Berry K. Differences in hepatocellular carcinoma risk, predictors and trends over time according to etiology of cirrhosis. PLoS One 2018;13, e0204412.
- [10] Estes C, Razavi H, Loomba R, Younossi Z, Sanyal AJ. Modeling the epidemic of nonalcoholic fatty liver disease demonstrates an exponential increase in burden of disease. Hepatology 2018;67:123–133.
- [11] Veterans Health Administration. Available at: http://www.va.gov/ health/findcare.asp [Last accessed 19 December 2016].
- [12] Veterans Affairs Corporate Data Warehouse. Available at: http://www. hsrd.research.va.gov/for\_researchers/vinci/cdw.cfm. [Last accessed on 19 December 2016].
- [13] Ioannou GN, Green PK, Berry K. HCV eradication induced by direct-acting antiviral agents reduces the risk of hepatocellular carcinoma. J Hepatol 2018;68:25–32.
- [14] Moon AM, Dominitz JA, Ioannou GN, Lowy E, Beste L. Use of antibiotics among patients with cirrhosis and upper gastrointestinal bleeding is associated with reduced mortality. Clin Gastroenterol Hepatol 2016.

- [15] Lerrigo R, Beste LA, Leipertz SL, Green PK, Lok AS, Kogut MJ, et al. Characteristics and outcomes of transjugular intrahepatic portosystemic shunt recipients in the VA Healthcare System. Eur J Gastroenterol Hepatol 2016;28:667–675.
- [16] Ioannou GN, Beste LA, Chang MF, Green PK, Lowy E, Tsui JI, et al. Effectiveness of sofosbuvir, ledipasvir/sofosbuvir, or paritaprevir/ritonavir/ombitasvir and dasabuvir regimens for treatment of patients with hepatitis C in the veterans affairs national health care system. Gastroenterology 2016;151:457–471, e5.
- [17] Ioannou GN, Bryson CL, Weiss NS, Boyko EJ. Associations between lipodystrophy or antiretroviral medications and cirrhosis in patients with HIV infection or HIV/HCV coinfection. Eur J Gastroenterol Hepatol 2015;27:577–584.
- [18] Ioannou GN, Bryson CL, Weiss NS, Miller R, Scott JD, Boyko EJ. The prevalence of cirrhosis and hepatocellular carcinoma in patients with human immunodeficiency virus infection. Hepatology 2013;57:249–257.
- [19] Beste LA, Ioannou GN, Larson MS, Chapko M, Dominitz JA. Predictors of early treatment discontinuation among patients with genotype 1 hepatitis C and implications for viral eradication. Clin Gastroenterol Hepatol 2010;8:972–978.
- [20] Kramer JR, Kanwal F, Richardson P, Giordano TP, Petersen LA, El-Serag HB. Importance of patient, provider, and facility predictors of hepatitis C virus treatment in veterans: a national study. Am J Gastroenterol 2011;106:483–491.
- [21] Backus LI, Boothroyd DB, Phillips BR, Mole LA. Predictors of response of US veterans to treatment for the hepatitis C virus. Hepatology 2007;46:37–47.
- [22] Davila JA, Henderson L, Kramer JR, Kanwal F, Richardson PA, Duan Z, et al. Utilization of surveillance for hepatocellular carcinoma among hepatitis C virus-infected veterans in the United States. Ann Intern Med 2011;154:85–93.
- [23] Kanwal F, Hoang T, Kramer JR, Asch SM, Goetz MB, Zeringue A, et al. Increasing prevalence of HCC and cirrhosis in patients with chronic hepatitis C virus Infection. Gastroenterology 2011;140:1182–1188, e1.
- [24] Miller DR, Safford MM, Pogach LM. Who has diabetes? Best estimates of diabetes prevalence in the Department of Veterans Affairs based on computerized patient data. Diabetes Care 2004;27(Suppl 2):B10–B21.
- [25] Kramer JR, Davila JA, Miller ED, Richardson P, Giordano TP, El-Serag HB. The validity of viral hepatitis and chronic liver disease diagnoses in Veterans Affairs administrative databases. Aliment Pharmacol Ther 2008;27:274–282.
- [26] Ascha MS, Hanouneh IA, Lopez R, Tamimi TA, Feldstein AF, Zein NN. The incidence and risk factors of hepatocellular carcinoma in patients with nonalcoholic steatohepatitis. Hepatology 2010;51:1972–1978.
- [27] Kramer JR, Giordano TP, Souchek J, Richardson P, Hwang LY, El-Serag HB. The effect of HIV coinfection on the risk of cirrhosis and hepatocellular carcinoma in U.S. veterans with hepatitis C. Am J Gastroenterol 2005;100:56–63.
- [28] Davila JA, Weston A, Smalley W, El-Serag HB. Utilization of screening for hepatocellular carcinoma in the United States. J Clin Gastroenterol 2007;41:777–782.
- [29] Beste LA, Green PK, Berry K, Kogut MJ, Allison SK, Ioannou GN. Effectiveness of hepatitis C antiviral treatment in a USA cohort of veteran patients with hepatocellular carcinoma. J Hepatol 2017;67:32–39.
- [30] El-Serag HB, Kanwal F, Richardson P, Kramer J. Risk of hepatocellular carcinoma after sustained virological response in Veterans with hepatitis C virus infection. Hepatology 2016;64:130–137.
- [31] White DL, Richardson P, Tayoub N, Davila JA, Kanwal F, El-Serag HB. The updated model: an adjusted serum alpha-fetoprotein-based algorithm for hepatocellular carcinoma detection with hepatitis C virus-related cirrhosis. Gastroenterology 2015;149:1986–1987.
- [32] El-Serag HB, Kanwal F, Davila JA, Kramer J, Richardson P. A new laboratory-based algorithm to predict development of hepatocellular carcinoma in patients with hepatitis C and cirrhosis. Gastroenterology 2014;146:1249–1255, e1.
- [33] Harrell Jr FE, Lee KL, Mark DB. Multivariable prognostic models: issues in developing models, evaluating assumptions and adequacy, and measuring and reducing errors. Stat Med 1996;15:361–387.
- [34] Gonen M, Heller G. Concordance probability and discriminatory power in proportional hazards regression. Biometrika 2005;92:1799–1809.
- [35] Royston P, Sauerbrei W. A new measure of prognostic separation in survival data. Stat Med 2004;23:723–748.
- [36] Miller ME, Langefeld CD, Tierney WM, Hui SL, McDonald CJ. Validation of probabilistic predictions. Med Decis Making 1993;13:49–58.

- [37] Graf E, Schmoor C, Sauerbrei W, Schumacher M. Assessment and comparison of prognostic classification schemes for survival data. Stat Med 1999;18:2529–2545.
- [38] Austin PC, van Klaveren D, Vergouwe Y, Nieboer D, Lee DS, Steyerberg EW. Geographic and temporal validity of prediction models: different approaches were useful to examine model performance. J Clin Epidemiol 2016;79:76–85.
- [39] Sharma SA, Kowgier M, Hansen BE, Brouwer WP, Maan R, Wong D, et al. Toronto HCC risk index: a validated scoring system to predict 10-year risk of HCC in patients with cirrhosis. J Hepatol 2018;68:92–99.
- [40] Vallet-Pichard A, Mallet V, Nalpas B, Verkarre V, Nalpas A, Dhalluin-Venier V, et al. FIB-4: an inexpensive and accurate marker of fibrosis in HCV infection. comparison with liver biopsy and fibrotest. Hepatology 2007;46:32–36.
- [41] Vickers AJ, Cronin AM, Elkin EB, Gonen M. Extensions to decision curve analysis, a novel method for evaluating diagnostic tests, prediction models and molecular markers. BMC Med Inform Decis Mak 2008;8:53.
- [42] Heller G. A measure of explained risk in the proportional hazards model. Biostatistics 2012;13:315–325.
- [43] Bruix J, Sherman M. Management of hepatocellular carcinoma. Hepatology 2005;42:1208–1236.
- [44] Davila JA, Morgan RO, Richardson PA, Du XL, McGlynn KA, El-Serag HB. Use of surveillance for hepatocellular carcinoma among patients with cirrhosis in the United States. Hepatology 2010;52:132–141.

- [45] Singal AG, El-Serag HB. Hepatocellular carcinoma from epidemiology to prevention: translating knowledge into practice. Clin Gastroenterol Hepatol 2015;13:2140–2151.
- [46] Goossens N, Singal AG, King LY, Andersson KL, Fuchs BC, Besa C, et al. Costeffectiveness of risk score-stratified hepatocellular carcinoma screening in patients with cirrhosis. Clin Transl Gastroenterol 2017;8, e101.
- [47] Atiq O, Tiro J, Yopp AC, Muffler A, Marrero JA, Parikh ND, et al. An assessment of benefits and harms of hepatocellular carcinoma surveillance in patients with cirrhosis. Hepatology 2017;65:1196–1205.
- [48] Ioannou GN, Green PK, Beste LA, Mun EJ, Kerr KF, Berry K. Development of models estimating the risk of hepatocellular carcinoma after antiviral treatment for hepatitis C. J Hepatol 2018;69:1088–1098.
- [49] Flemming JA, Yang JD, Vittinghoff E, Kim WR, Terrault NA. Risk prediction of hepatocellular carcinoma in patients with cirrhosis: the ADRESS-HCC risk model. Cancer 2014;120:3485–3493.
- [50] Bruix J, Sherman M. AASLD Practice Guideline. Management of Hepatocellular Carcinoma: An Update. Available at: http://www.aasld.org/sites/ default/files/guideline\_documents/HCCUpdate2010.pdf. 2011.
- [51] Kassirer JP, Moskowitz AJ, Lau J, Pauker SG. Decision analysis: a progress report. Ann Intern Med 1987;106:275–291.
- [52] Raji OY, Duffy SW, Agbaje OF, Baker SG, Christiani DC, Cassidy A, et al. Predictive accuracy of the Liverpool Lung Project risk model for stratifying patients for computed tomography screening for lung cancer: a casecontrol and cohort validation study. Ann Intern Med 2012;157:242–250.